Author’s response to reviews

Title: TIPIT: A randomised controlled trial of thyroxine in preterm infants under 28 weeks gestation: Magnetic Resonance Imaging and Magnetic Resonance Angiography protocol

Authors:

Sze M Ng (M.Ng@liverpool.ac.uk)
Mark A Turner (mark.turner@liverpool.ac.uk)
Carrol Gamble (C.gamble@liverpool.ac.uk)
Mohammed Didi (mohammed.didi@rlc.nhs.uk)
Suresh Victor (Suresh.victor@manchester.ac.uk)
Christina Malamateniou (Christina.Malamateniou@liverpool.ac.uk)
Laura M Parkes (laupar@liverpool.ac.uk)
Anna Tietze (anna.tietze@rlc.nhs.uk)
Lloyd Gregory (lloyd.gregory@manchester.ac.uk)
Vanessa Sluming (vanessa.sluming@liverpool.ac.uk)
Laurence Abernethy (laurence.abernethy@rlc.nhs.uk)
Alan M Weindling (a.m.weindling@liverpool.ac.uk)

Version: 7 Date: 28 June 2008

Author’s response to reviews:

Formatting changes requested

-----------------------------

Title - Please use sentence case in the title (capitalise only the first word, and proper nouns), both in the manuscript file and the submission system (i.e. remove all unnecessary capitals).

ACTION: This is ammended

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

ACTION: This is ammended

References - References should be cited in the text using consecutive numbers in brackets. Please ensure the type of bracket used is consistent by using square brackets [ ]. Please do not use superscript numbers. Also, please place the reference before the full stop.

ACTION: This is ammended

Tables must be created using a Tables tool. Tables must be divided into rows
and columns and be oriented in a portrait page layout. The borders of tables included, as part of the manuscript file, must be visible black lines. Tables generated with tabbed text are not acceptable. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. We are unable to correctly display merged cells where the merged cell crosses rows.

We also prefer tables and figures to fit onto an portrait A4 page. If possible please re-orientate your table to the portrait orientation.

Table in parts - Table 2 has been divided into 2.1, 2.2 and 2.3. It is important that tables are numbered in ascending numerical order - 1, 2 and 3. Can we please ask that the tables are renumbered and any references to the tables in the text are updated to reflect these changes.

ACTION: This is ammended and all tables are in simple style and in portrait layout. References to Table 2 are ammended in the text.

Website addresses - Websites must be removed from the text, and cited from the main reference list only. If you are using referencing software such as EndNote or Refman, please ensure that your references are entered through the software rather than included in the manuscript reference list manually. Please present your reference in the following structure, e.g.

17. Website title [www.website.com].

This section is ammended as the details are described in reference 53.